Author’s response to reviews

Title: A case report of a mild form of multiple acyl-CoA dehydrogenase deficiency due to compound heterozygous mutations in the ETFA gene

Authors:
Robin Chautard (robin.chautard@chu-limoges.fr)
Cécile Laroche-Raynaud (cecile.laroche-raynaud@chu-limoges.fr)
Anne-Sophie Lia (anne-sophie.lia@unilim.fr)
Pauline Chazelas (pauline.chazelas@gmail.com)
Paco Derouault (paco.deroault@chu-limoges.fr)
Franck Sturtz (franck.sturtz@unilim.fr)
Yasser Baaj (yasser.baaj@chu-limoges.fr)
Alice Veauville-Merllié (alice.veauville-merllie@chu-lyon.fr)
Cécile Acquaviva (cecile.acquaviva-bourdain@chu-lyon.fr)
Frédéric FAVREAU (frederic.favreau@chu-limoges.fr)
Pierre-Antoine FAYE (pierreantoine.faye@gmail.com;pierre-antoine.faye@chu-limoges.fr;pierreantoine.faye@unilim.fr)

Version: 2 Date: 09 Jan 2020

Author’s response to reviews:
Response to Editor and Reviewers

Dear editor,

Please find enclosed the revision of our manuscript entitled “A case report of a mild form of multiple acyl-CoA dehydrogenase deficiency due to compound heterozygous mutations in the ETFA gene”. The manuscript was edited by a reputable English language editing service.
We are grateful for your time and your expertise.

Sincerely yours,

Pierre-Antoine FAYE

Editor comments:

"The manuscript is acceptable, but reviewers have suggested editing by a native English speaker. Please see instructions below for information. Please also run a spell-check prior to final submission (e.g., pg. 6: deshydrogenase)."

Thank you for pointing out this point. We checked ran a spell-check prior the final submission. The manuscript was edited by William Hempel an Alex Edelman & associates employee (http://www.alexedelman.com/).

Jerry Vockley (Reviewer 1):

“This is a revised manuscript describing a patient with MADD and new variant identified in the ETFA gene. The authors have addressed all of my scientific critiques. Some of the language is still a bit clumsy and the manuscript could stand to be edited by a native English speaker.”

Thanks a lot for your comment. The manuscript was edited by William Hempel an Alex Edelman & associates employee (http://www.alexedelman.com/).
Brian Meyer (Reviewer 2):
[No comments provided]

Celia Nogueira (Reviewer 3):
Review accepted